**Author’s response to reviews**

**Title:** A pilot feasibility randomised controlled trial of an adjunct brief social network intervention in opiate substitution treatment services in England

**Authors:**

Edward Day (edward.day@kcl.ac.uk)
Alex Copello (a.g.copello@bham.ac.uk)
Jennifer Seddon (Jennifer.Seddon@hotmail.com)
Marilyn Christie (marilynmchristie@gmail.com)
Deborah Bamber (Deborah.Bamber@dmu.ac.uk)
Charlotte Powell (Charlotte.Powell@dmu.ac.uk)
Carmel Bennett (c.bennett.1@pgr.bham.ac.uk)
Shabana Akhtar (Shabana.Akhtar@bsmhft.nhs.uk)
Sanju George (sanjugeorge531@gmail.com)
Andrew Ball (andrew.ball@turning-point.co.uk)
Emma Frew (e.frew@bham.ac.uk)
Ilias Goranitis (i.goranitis@bham.ac.uk)
Nicholas Freemantle (Nicholas.Freemantle@ucl.ac.uk)

**Version:** 2  **Date:** 10 Dec 2017

**Author’s response to reviews:**

1. Please relocate all relevant trial registration information so that it appears at the end of the Abstract section.

All registration information has been moved to the end of the Abstract section.
2. As indicated by our submission guidelines, please include a Conclusions heading in the text of your manuscript, as both a Conclusion and a Discussion section should be provided. You may consider placing it before the last paragraph of the Discussion section.

A separate Conclusions section has been added to the end of the Discussion section.

3. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

A full list of abbreviations has been added after the Conclusions section.

4. Under your Ethics approval heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal.

Informed consent was always written, and this has been clarified in the Declarations section.

5. Please add a Consent for publication heading to the Declarations section. Under this heading you may state “Not applicable” as no personal information is provided in your manuscript.

A Consent for Publication heading has been added, and the ‘not applicable’ response added.

6. Please include an Acknowledgements subheading in your Declarations section. In this section acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Please note that any sections that are not applicable to your manuscript should still be included with the statement “Not applicable.”

An Acknowledgements section has been added.
7. Thank you for including a Funding heading for the Declarations section. Please also state the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The role of the funding body has been clarified in the Declarations section

8. Please note that while smaller tables may be inserted into the text or pasted at the end of the manuscript file after references, larger tables should be placed after the References section. Therefore, please move table 2 accordingly

Table 2 has been moved to the end of the paper, after the References section

9. Please remove any figure titles and captions from the figure files as these files should contain the image graphics only. Upon doing this, please place the titles/captions at the end of the main manuscript after the References section under a newly created “Figure legend” heading.

The figure title has been removed from the figure file and added as part of a new Figure Legend section after the References

10. Please add a “Supplementary files” section where you list the following information about your supplementary material:

A Supplementary files section has been added at the end of the paper

11. We note that the current submission contains some textual overlap with other previously published works. This overlap mainly exists under the Background (lines 33-50, pages 3-4) and Methods (lines 42-2, pages 5-12) sections.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase these sections to minimise overlap.
The Background section and the Interventions, Staff Recruitment & Training, and Participant Recruitment & Randomisation sections of the Methods have been re-written to reduce the similarity to the previous protocol paper.